Author's response to reviews

Title: Subacute and chronic, non-specific back and neck pain: cognitive-behavioural rehabilitation versus primary care. A randomized controlled trial.

Authors:

Odd Lindell (odd.lindell@ki.se)
Sven-Erik Johansson (sven-erik.johansson@ki.se)
Lars-Erik Strender (lars-erik.strender@ki.se)

Version: 5 Date: 19 December 2008

Author's response to reviews:

Dear member of The BioMed Central Editorial Production Team,

Out from your valuable correction suggestions I have made the following revisions:
- All tables are now created using a tables tool.
- Table 6 in converted to a portrait page layout.

Best regards,

Odd Lindell
Center for Family and Community Medicine,
Karolinska Institutet, Alfred Nobels allé 12,
SE-141 83 Huddinge, Sweden
phone +46-8-524 800 00, fax +46-8-524 887 07